Author's response to reviews

Title: One stage surgical treatment of aortic valve disease and aortic coarctation with aortic bypass grafting through the diaphragm and aortic valve replacement

Authors:

Zipu Yu (yuzipuleaf@sina.com)
Shengjun Wu (928115034@qq.com)
Chengchen Li (269158070@qq.com)
Yu Zou (2236504826@qq.com)
Liang Ma (maliang99@yahoo.com)

Version: 3
Date: 5 September 2015

Author's response to reviews: see over
Dear editors and reviewer:

On behalf of my co-authors, we appreciate editors and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “One stage surgical treatment of aortic valve disease and aortic coarctation with aortic bypass grafting through the diaphragm and aortic valve replacement”. We have studied reviewer’s comments carefully. Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have made revision which marked in red in the paper. We have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration. The responds to the reviewer’s comments are as following:

Responds to the reviewer’s comments:

Reviewer 2:

1. Response to comment: The manuscript needs minor revisions of the language.

Response: We have made revision according to the Reviewer’s comment, which marked in red in the paper.

Reviewer 1:

1. Response to comment: Figure 1 A could be removed. Figure 2C could be abolished.

Response: Considering the Reviewer’s suggestion, we deleted Figure 1 A while reserved Figure 2C. The reason for our action is that Figure 2C shows the procedure of end-to-side graft-to ascending aorta anastomosis.

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper. And here we did not list the
changes but marked in red in revised paper. We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for your comments and suggestions. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Zipu Yu, Shengjun Wu, Chengchen Li, Yu Zou, Liang Ma

Corresponding author:

Name: Liang Ma

E-mail: maliang99@yahoo.com